

JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2020JG006120

Key Points:

- Both conventional linear and microbial nonlinear models can capture the pulsed dynamics of heterotrophic respiration
- The moisture-response function combined with model parameterization dominates the modeled uncertainties in heterotrophic respiration
- More observations of nonlinear C phenomena are needed to reduce the simulation uncertainty on soil C cycling in Earth system models

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

J. Xia and X. Zhang, jyxia@des.ecnu.edu.cn; xuanzezhang@igsnrr.ac.cn

Citation:

Zhou, J., Chen, S., Yan, L., Wang, J., Jiang, M., Liang, J., et al. (2021). A comparison of linear conventional and nonlinear microbial models for simulating pulse dynamics of soil heterotrophic respiration in a semi-arid grassland. *Journal of Geophysical Research: Biogeosciences*, *126*, e2020JG006120. https://doi. org/10.1029/2020JG006120

Received 19 OCT 2020 Accepted 12 APR 2021

A Comparison of Linear Conventional and Nonlinear Microbial Models for Simulating Pulse Dynamics of Soil Heterotrophic Respiration in a Semi-Arid Grassland

Jian Zhou^{1,2}, Shiping Chen³, Liming Yan^{1,2}, Jing Wang^{1,2}, Ming Jiang^{1,2}, Junyi Liang⁴, Xuanze Zhang^{5,1}, and Jianyang Xia^{1,2}

¹Zhejiang Tiantong Forest Ecosystem National Observation and Research Station, State Key Laboratory of Estuarine and Coastal Research, School of Ecological and Environmental Sciences, East China Normal University, Shanghai, China, ²Research Center for Global Change and Ecological Forecasting, School of Ecological and Environmental Sciences, East China Normal University, Shanghai, China, ³Institute of Botany, Chinese Academy of Sciences, Beijing, China, ⁴Department of Grassland Resource and Ecology, College of Grassland Science and Technology, China Agricultural University, Beijing, China, ⁵Institute of Geographic Sciences and Natural Resources Research, Chinese Academy of Sciences, Beijing, China

Abstract As a critical process in regulating terrestrial feedback to climate change, soil heterotrophic respiration is commonly simulated with the first-order kinetics in current Earth system models. Compared with the first-order kinetic models, explicit microbial models are expected to better simulate nonlinear carbon (C)-cycle phenomena, such as the pulse dynamics of soil heterotrophic respiration driven by dry-rewetting cycles in grasslands. However, these two types of models (i.e., linear conventional and nonlinear microbial models) have never been compared based on in situ observations of soil heterotrophic respiration in the semi-arid grassland, which is significantly affected by the dryrewetting events. Here, based on the field data of soil heterotrophic respiration in a semi-arid grassland in northern China, we first showed that the shift from a conventional linear model to a nonlinear microbial model did not substantially improve the simulation of soil heterotrophic respiration. Then, we quantified the contributions of different moisture-response functions to the uncertainty in simulating the soil C dynamics. The results showed that the selection of moisture-response functions combined with parameterization in the soil C models dominated the modeled uncertainties in soil heterotrophic respiration. These findings suggest that both the conventional linear model and nonlinear microbial model can simulate well the pulse dynamic of soil heterotrophic respiration in grasslands with an improved parameterization of water regulation on soil carbon decomposition. This study also calls for more observations of nonlinear C phenomena for reducing the simulation uncertainty on soil C cycling in Earth system models.

Plain Language Summary Soil heterotrophic respiration refers to the process of decomposing soil organic carbon and releasing CO₂ into the atmosphere. Current soil carbon models use the first-order kinetics to represent the decomposition process, but large uncertainties have emerged in simulating soil heterotrophic respiration among these models. Therefore, models incorporating microbial mechanisms are expected to better simulate soil carbon processes than the conventional models, especially in simulating the nonlinear phenomena of soil heterotrophic respiration. To test the ability of these two types of models (i.e., linear conventional and nonlinear microbial models) in simulating the nonlinear soil processes, we evaluate them based on in situ observations of the pulsed dynamics of soil heterotrophic respiration in a semi-arid grassland. We found the microbial-explicit model did not substantially improve the simulation of soil heterotrophic respiration. The moisture-response function combined with parameterization played an important role in reducing the uncertainties in simulating the pulsed dynamics of soil heterotrophic respiration in both types of models.

1. Introduction

The dynamic of soil organic carbon (C) under climate change is a key regulator of ecosystem C cycling due to the large C storage capacity in the soil and its high sensitivity to environmental changes (Bradford et al., 2016; Davidson & Janssens, 2006; Doetterl et al., 2015). To improve our understanding of the soil C

All Rights Reserved.

© 2021. American Geophysical Union.



dynamics and its interactions with the past and future climate changes, many numerical soil C-cycle models based on the first-order kinetics have been developed. The decomposition of soil organic matters in those models, for example, CENTURY (Parton et al., 1987) and RothC (Jenkinson et al., 1991), is typically represented as multiple pools with first-order decay rates. Although the first-order linear multi pool models have been widely used in Earth system modeling, large uncertainties have emerged in simulating soil C dynamics and soil heterotrophic respiration (R_H) among models because of their differences in model structure and parameterization (Bradford et al., 2016; Todd-Brown et al., 2013; Zhou et al., 2021).

One promising approach to reduce the large uncertainties on soil C dynamics in Earth system models is explicitly representing microbial processes with the multi pool models (Luo et al., 2016; Wieder, Allison, et al., 2015). Some recent efforts have been made to develop nonlinear soil C models based on microbial physiological processes (Abramoff et al., 2018; Allison et al., 2010; German et al., 2012; Wang et al., 2013; Wieder et al., 2013; Wieder, Grandy, et al. 2015). In many examples, the microbial process-based models have improved the modeling of the soil C dynamic and its feedbacks to climate change (Wang et al., 2013, 2016; Wieder, Allison, et al., 2015). However, it should be noted that a few other studies also have shown that current microbial models could conversely amplify the uncertainty in predicting soil C responses to climate change due to the complex mechanisms in microbial processes and challenges in model parameterization (Shi et al., 2018; Sulman et al., 2018).

Another important way to improve the accuracy of soil C dynamics in Earth system models is better parameterizations of environmental impacts in the model (Falloon et al., 2011; Sierra et al., 2015; Xia et al., 2013). The environmental response function in current linear or nonlinear soil C models varies significantly among models and hence causes considerable uncertainty in simulating soil C dynamics (Falloon et al., 2011; Sierra et al., 2015). For example, in a grassland ecosystem, Lei et al. (2018) have indicated that different water scalars greatly affected the modeled ecosystem respiration and can result in a large spread of the future projections of the ecosystem C sink.

The above activities of model improvement lead to a new question of how linear conventional and nonlinear microbial models will perform if they are used to simulate the nonlinear C-cycle phenomena, such as the rain-pulse effect in semi-arid grasslands. The rain-pulse effect, often referred to as the "Birch effect," is a typical microbially driven nonlinear soil process (Birch, 1964; Lawrence et al., 2009; Schimel & Weintraub, 2003). The rain-pulse effect of R_H is an essential contributor to soil-to-air C (e.g., CO_2 , CH_4) release in semi-arid grasslands (Yan et al., 2014). The pulse C release can be 30-times higher than under normal hydrological conditions and contribute to as much as 40% of ecosystem CO_2 -efflux in some grasslands (Sponseller, 2007; Yan et al., 2014). Accurate simulation of the rainfall pulse effect is relevant for understanding the soil C dynamics of semi-arid grasslands. However, modeling studies have shown that the rain-pulse effect of R_H can be captured by some soil microbial models (Salazar et al., 2018; Schimel & Weintraub, 2003; Waring & Powers, 2016) or by some conventional linear models (Waring & Powers, 2016). Therefore, the pulse response of R_H to rainfall events in semi-arid grasslands provides an ideal case for evaluating the performance of different soil C models.

In this study, we applied eight terrestrial C cycling models by incorporating two different soil C model structures (linear conventional or nonlinear microbial) and four soil water-scalar schemes. Using these models, we conducted numerical experiments to simulate the rain pulse effect on soil C dynamics in a semi-arid grassland. Based on the comparative analysis, this study aimed to evaluate (1) whether the conventional linear model can capture nonlinear soil C process due to the rain-pulse effect as microbial models do, and (2) how the moisture response function affects the model performance in simulating the nonlinear soil C process.

2. Materials and Methods

2.1. Site Description and Data Sources

The observational soil CO₂-efflux data and meteorological forcings for models were obtained during the growing season in 2009 from a continuous field manipulative experiment in a semi-arid grassland in China. The study site is located at Duolun County (42°27′N, 116°41′E) in northeastern Inner Mongolia, China. The dominant species of this region are *Stipa krylovii*, *Agropyron cristatum*, *Leymus chinensis*, and *Artemisia*



frigida. According to the U.S. soil taxonomy classification system (Yuan et al., 2005), the soil at the study site is classified as Calcic-orthic Aridisol. The mean annual air temperature is 3° C, ranging from -30° C in the winter to 34° C in the summer. The mean annual precipitation is 377 mm, nearly 95% (i.e., 358 mm) of which fall in the growing season (May to October).

The R_H was measured based on the clipping method (Byrne & Kiely, 2006). This method was used to separate the auto- and heterotrophic parts of soil respiration, which assumed that continuous removal of living plants greatly inhibited root growth in the clipped plots and the soil CO_2 -efflux in the plots would mainly come from soil fauna and microbial respiration. Therefore, three 1×1 m clipped plots 5–30 m apart were designed in late April 2006. The living plants in these clipped plots and an area of 10 cm around the plots were removed a month before the first measurement. Meanwhile, a PVC collar (20.3 cm in diameter and 8 cm in height) was inserted into the soil at a depth of 3 cm in each plot one month before the first measurement. Soil CO_2 efflux was continuously measured every half an hour during the growing season using an infra-red gas analyzer (LI-840, Li-Cor Inc., Lincoln, NE, USA) with three automatic measurement chambers. Soil CO_2 efflux rate was determined by the time-series of CO_2 concentrations in these chambers, which were recorded by a CR1000 data logger (Campbell Scientific Inc., CSI, USA Utah, Utah) and were processed by LoggerNet 3.1 (CSI, USA). Each measurement lasted 120 s. This measurement lasted from the beginning of May to the end of October during 2009. Most missing data gaps were less than 2 h and were filled by linear interpolation based on the nearby data or the linear relationship between R_H and soil moisture.

The precipitation and the CO_2 fluxes, including net ecosystem exchange (NEE) and ecosystem respiration (ER), were recorded at a half-hour time step by a nearby eddy covariance flux tower located 4 m above the ground. In this study, NEE represented the net CO_2 exchange between ecosystem and atmosphere, and a positive NEE means ecosystem carbon release while the negative NEE means ecosystem carbon uptake. The gross primary productivity (GPP) was calculated as the difference between NEE and ER. More details of the measurement and calculation method have been described in Zhang et al. (2007). Soil temperatures were measured by soil temperature probes (CSI, USA) with a depth of 10 cm. The volumetric soil moistures were measured by the CS616 soil water probes (CSI, USA) at a depth of 0–10 cm. These probes were placed near the soil CO_2 efflux chambers and the mean values of half-hour soil temperature and moisture were recorded simultaneously in the CR1000 data logger (CSI, USA).

2.2. Model Description

2.2.1. Model Structures

To evaluate how well different model structures capture the soil C dynamics in response to the rain events, we incorporated two models (linear conventional vs. nonlinear microbial model) into a terrestrial C cycling model that is a simplified version of the Terrestrial ECOsystem (TECO) model (Weng & Luo, 2008). The conventional linear model is based on Liang et al. (2018), while the nonlinear microbial model is based on Allison et al. (2010), outlined in Figure 1. Both models include the same above ground C allocation process and the input of litterfall. We used the measured GPP as the input of the models.

The conventional model includes three soil pools divided based on their decomposition rates: a fast soil C, a slow soil C, and a passive soil C (Figure 1a). Each pool in the conventional model has a constant turnover rate, and CO_2 is released from each pool through C decomposition, and a fraction of C in one pool is transferred to another. Parameters describing fractions between the three soil C pools are shown in Table S1. The decomposition of each C pool is represented by the following equation (Lawrence et al., 2009):

$$\frac{dC_i}{dt} = k_i C_i M_s T_s \tag{1}$$

where C_i is the pool size (g C m⁻²) and k_i is the turnover rate in each specific pool (day⁻¹). T_s and M_s are scalar schemes of temperature response function and moisture response function (MRF) respectively to modify the C decomposition rate depending on the abiotic environment. The temperature response function (T_s) in all models of this study was expressed as follows (Lei et al., 2018):





Figure 1. The diagram of the two carbon cycle models in this study, (a) with conventional structure (from TECO model) and (b) microbial structure (Allison et al., 2010) in the part of the soil, respectively (Fast soil carbon pool, Fast; Slow soil carbon pool, Slow; Passive soil carbon pool, Passive; Soil organic carbon, SOC; Dissolve organic carbon, DOC; Microbial biomass carbon, MBC; Enzyme biomass carbon, EBC). TECO, Terrestrial ECOsystem.

$$T_s = Q_{10}^{\frac{T_{\text{soil}} - T_{\text{ref}}}{T_{\text{ref}}}}$$
(2)

where the T_{soil} represents the soil temperature (°C). Q_{10} and T_{ref} are set to 2 and 10 (°C), respectively (Friedlingstein et al., 2006).

The microbial soil C cycling model used in this study incorporates four pools: a microbial biomass carbon (MBC), a dissolved organic carbon (DOC), a soil enzyme biomass carbon (EBC), and a soil organic carbon (SOC) (Figure 1b). The two processes of the decomposition of SOC and the assimilation of MBC from the DOC pool are represented as the Michaelis-Menten function (Allison et al., 2010):

$$F_s = V_{\max} \times \text{EBC} \times \frac{\text{SOC}}{\text{Km} + \text{SOC}}$$
(3)

$$F_U = V_{\text{maxuptake}} \times \text{MBC} \times \frac{\text{DOC}}{\text{Km}_{\text{uptake}} + \text{DOC}}$$
(4)

where F_S is the fraction of SOC decomposition. F_U is the fraction of MBC uptaking from DOC. V_{max} and $V_{\text{max uptake}}$ indicate the maximal decomposition rate (day⁻¹), which is calculated by the Arrhenius equation. Km and Km_{uptake} (g C m⁻²) is the half-saturation of the MBC or EBC on the substrate. Microbial death and enzyme production are modeled as a constant fraction of MBC. The enzyme turnover is a first-order process with a constant rate. More details about the parameters and processes are described in Allison et al. (2010).

Heterotrophic CO_2 (R_H) is the fraction of DOC assimilated by microbes that is not allocated to biomass production:

$$R_{H} = V \max_{\text{uptake}} \times \text{MBC} \times \frac{\text{DOC}}{\text{Km}_{\text{uptake}} + \text{DOC}} (1 - \text{CUE})$$
(5)

where $V_{max_{uptake}}$ represents the maximum decomposition rate of DOC assimilated by microbes. MBC is the microbial biomass carbon. Km_{uptake} is the half-saturation of the substrate. CUE is the carbon use efficiency.

More descriptions of parameters for the microbial model are shown in Table S2. More details of the microbial model can be found in Allison et al. (2010), such as the changes of MBC and EBC.

Considering the effects of environmental factors (e.g., temperature and moisture), we added the temperature and water responses into the Michaelis-Menten function:



Table 1

Moisture Response Function (MRF) Parameterizations on Soil Carbon Decomposition From Four Different Terrestrial Biosphere Models

	Provenance	Moisture response function
MRF ₁	CABLE	$\left(\frac{1.7 - W_C}{1.15}\right)^{6.6481} \left(\frac{W_C + 0.007}{0.557}\right)^{3.22}$
MRF ₂	IBIS	$e^{rac{-\left(w_C - 0.6 ight)^2}{0.08}}$
MRF ₃	LPJ-DGVM	$0.25 + 0.75 W_C$
MRF ₄	TECO	$\begin{cases} 5 \times WC \ (W_C < 0.2) \\ 1 \ (W_C \ge 0.2) \end{cases}$

Note. WC: soil water content (v/v%).

Abbreviation: TECO, Terrestrial ECOsystem.

$$\frac{dC_i}{dt} = V_{\max} \times C_i \times \frac{S}{K+S} M_s T_s \tag{6}$$

where the M_s and the T_s are the scalar schemes as used in the linear soil C model.

2.2.2. Soil Water Response Parameterizations

To assess and compare the contribution of different MRF schemes (i.e., M_s) to the uncertainty in the soil C dynamics modeling, we incorporated four soil MRF schemes in the TECO model and re-parameterized them (Table 1). These MRFs are all derived from terrestrial C cycle models. The scheme MRF₁ from the CABLE model (Wang et al., 2010) and the MRF₂ from IBIS model (Kucharik et al., 2000) are nonlinear functions of soil moisture, while the MRF₃ from LPJ-DGVM model (Sitch et al., 2003) and the MRF₄ from TECO model (Weng & Luo, 2008) are linear functions of soil moisture. Their response curves to soil moisture are shown in Figure S1. The moisture-response functions MRF₁ and MRF₂ start from 0

and grow to unit 1 when the soil water content exceeds 50%. While the linear response function MRF_3 changes from 0.25 to 0.6 with soil water content from 0% to 50%. The linear response function MRF₄ has the maximum growth rate with a range from 0 to 1 when the soil water content increases from 0% to 20%.

2.3. Data Assimilation for Improving Model Parameterization

In this study, we combined two C-cycle structures (conventional and microbial) models with four MRF schemes, which then produce eight model variants. All of them had parameter optimizations via a data assimilation method before their comparisons. To optimize the parameters of models, the conditional Bayesian inversion was used to assimilate data in this study (Du et al., 2017; Jiang et al., 2018; Li et al., 2016; Ma et al., 2017). The Bayes' theorem can be expressed as follow:

$$p(\theta \mid Z) \propto p(Z \mid \theta) P(\theta)$$
(7)

where $P(\theta)$ represents the prior knowledge of the parameters in the models, referring to the previous studies (Allison et al., 2010; Shi et al., 2018). The detailed description of the parameters is shown in Tables S1 and S2. Additionally, the likelihood function $p(Z \mid \theta)$ was calculated based on the assumption that the errors (with mean biases removed) between each observation data and model simulation results followed a Gaussian distribution with a zero mean. It can be expressed as follows:

$$p(Z \mid \theta) \propto \exp\left\{-\sum_{t \in Z} \frac{\left[Z(t) - X(t)\right]^2}{2\sigma^2(t)}\right\}$$
(8)

where Z(t) is the observation stream at time t, X(t) is the corresponding simulated variable, and $\sigma(t)$ is the standard deviation of the observation set.

The posterior probability distribution of parameters was obtained by the Markov chain Monte Carlo (MCMC) technique with adaptive Metropolis-Hastings (M-H) algorithm (Hastings, 1970; Metropolis et al., 1953). The new proposal parameters were generated by:

$$C^{\text{new}} = C^{k-1} + r \left(\theta_{\text{max}} - \theta_{\text{min}} \right)$$
(9)

where θ_{\max} and θ_{\min} are the maximum and minimum values of the given parameter space, respectively (shown in Tables S1 and S2), C^{k-1} means the parameter set of the previous step, and r is a random variable



between -0.5 and 0.5 with a uniform distribution. The initial samples (approximately 5,000 for each run) were discarded after the running means, and standard deviations (SDs) were stabilized (regarded as the burn-in period) (Du et al., 2017). All the accepted samples without the burn-in periods were used for posterior analysis. In total, there are 50,000 accepted samples from five parallel chains to construct the posterior distribution.

The initial values of total soil C used in the models were obtained from measurements (Liu et al., 2013). The optimized values of the ratio of each C pool to the measured total soil C in the conventional model are used as the initial stable state of the model. The upper ranges of the initial values of MBC, EBC, and DOC in the microbial model for data assimilation were set $\sim 1\%$ of total C carbon (Wieder et al., 2013). To equilibrate the models under different parameter sets, we set each step of data assimilation to run 7,300 time-steps before the start of the simulations used to examine if the new parameter set should be accepted or rejected. The detailed information on the initial state variables in the models is shown in Table S3.

2.4. Model Evaluation

To evaluate the ability of models to simulate the soil C dynamics, we compared the modeled R_H with experimental data by deviance information criterion (DIC) (Liang et al., 2018; Spiegelhalter et al., 2002). In our study, half of the observation data were randomly selected for parameter optimization by data assimilation and the other half for verification. For each model, DIC was calculated by

$$DIC = \overline{D} + p_D \tag{10}$$

where

$$\overline{D} = \frac{1}{n} \sum_{t=1}^{n} \left(-2 \log \left(p(Z \mid \theta^{t}) \right) \right)$$
(11)

and

$$p_D = \overline{D} + 2\log\left(p(Z \mid \overline{\theta})\right) \tag{12}$$

where *n* is the number of the generated parameter sets, and $\overline{\theta}$ is the mean of the generated parameter sets. A smaller DIC for a model means a better simulation against observational data.

We randomly selected 1,000 parameter sets from the posterior probability distribution formed by data assimilation in each model from the eight model combinations. We further obtained the DICs of the simulations from these parameter sets. The DIC results were used to evaluate the performance of these models. Comparing the DIC results of conventional and microbial models, we analyzed which of these model structures can better capture the rain pulse.

MRF can affect the decomposition of different soil C pools and thus affect the performances of models in simulating R_H . In our study, the relationships between R_H and the changes of different soil C pools are measured by the partial correlation that is conducted by the software of Matlab (R2019a). Meanwhile, different moisture-response functions can change the sensitivity of simulating R_H to the parameters in the model. The sensitivity of parameters in our study is determined by the sensitivity index (I) defined as (Lenhart et al., 2002):

$$I = \frac{\left| \left(y_2 - y_1 \right) \right| / y_0}{2\Delta x / x_0}$$
(13)

Where y_0 is the model output (R_H) with the initial parameter set x_0 . The model outputs from initial parameters varied by $\pm \Delta x$ are variable y_2 and y_1 . Δx was set at 0.25 times of initial values. Details of the sensitivity index are available in Lenhart et al. (2002).





Figure 2. Time-series of observed R_H (red line), soil moisture (black line), precipitation (black rectangle) (a), and the scalar ability of four different moisture response functions (MRF) under the conditions of observed soil moisture (b).

3. Results

3.1. Observed Rain Pulse in a Grassland

From the observation-based analysis, we show that the patterns of soil moisture and R_H were significantly affected by the rainfall events during the growing season (Figure 2a). Generally, rainfall events varied from 0.3 to 33.8 mm, and the observed soil moisture increased by 0.4%–132% accompanied by rainfall events. The observed R_H also showed obvious variations and increased by 4.4%–369.2% in 1–2 days after the rainfall events. After reaching its peak, R_H gradually returned to pre-rainfall levels in 2–5 weeks (Figure 2a). Additionally, MRFs used in our study showed different regulation abilities under the measured soil moisture conditions. As shown in Figure 2b, during the growing season, the water response scalar of MRF₁ and MRF₂ changed from 0.0058 to 0.12 and 0.022 to 0.079, respectively, while MRF₃ and MRF₄ ranged from 0.28 to 0.36 and 0.23 to 0.75, respectively (Figure 2b).

3.2. Model Validation and Comparison

The data assimilation method could effectively constrain most of the targeted parameters in all models (Figure S2). After optimizing the parameters by conditional Bayesian method (see Tables S1 and S2 for optimized parameters in conventional and microbial models, respectively), the modeled daily R_H from May to October in 2009 matched more closely to the observed values both on timing and magnitude, with differences among models (Figure 3). Most models exhibited similar bias in simulating R_H during the growing season, which showed an underestimation during the rewetting periods and an overestimation during the dry periods, while models with the MRF₁ scheme showed that a slight overestimation during rewetting periods and an underestimation during dry periods (Figure 3).

We further evaluated the performance of the eight models. The values of DIC varied greatly among different MRFs in each model structure (Figure 4). Generally, the $R_{\rm H}$ from microbial models showed higher DICs





Figure 3. Time-series of observed and simulated daily $R_{\rm H}$ in Inner Mongolia grassland from May to October in 2009. MRF₁-MRF₄ represent moisture response function from CABLE, IBIS, LPJ, and TECO, respectively. (a, c, e, and g) represent conventional models with four different moisture response functions, respectively. (b, d, f, and h) represent microbial models with four moisture response functions, respectively (Blue and red range for simulation from conventional and microbial models with posterior parameter sets and solid line for models with parameters of maximum likelihood. Block line and range are observations). TECO, Terrestrial ECOsystem.









Figure 5. Time-series of the contributions of different soil carbon pools to soil heterotrophic respiration (R_H) in conventional models (MRF_1-MRF_4 are moisture response functions from the model of CABLE, IBIS, LPJ, and TECO, respectively). TECO, Terrestrial ECOsystem.

(ranged from 76 to 156) and larger variances than those in conventional models (ranged from 57 to 140) when using the same water scheme (Figure 4). Specifically, the differences between the DIC values for conventional and microbial models were the largest using MRF_2 or MRF_4 schemes. This result indicated that the conventional models show better performance than the microbial models using the same MRF schemes. The conventional model using the MRF_2 scheme (the water response function from IBIS) had the lowest DIC score, showing the best performance among all models (Figure 4).

3.3. Impact of the Moisture Response Function

The moisture-response function mainly regulated the decomposition rate of soil C pools in the model. We found that in conventional models with MRF_1 or MRF_2 , the R_H mainly came from the decomposition of fast soil C, contributing more than 95% of R_H , while for MRF_3 and MRF_4 , the fast and slow soil C pools contribute almost equally to the release of CO_2 efflux as R_H from the soil C (Figure 5). The passive C pool in all conventional models contributed negligibly at the seasonal time scale in 2009 (Figure 5).

In the microbial models, the CO₂ production from soil C was directly from the process of microbial absorbing C from DOC and indirectly affected by SOC and EBC that affected the size of MBC and DOC. The cumulative changes in SOC, MBC, DOC, and EBC during the whole growing season were about -200 ± 29.5 , 27 ± 3.8 , -0.1 ± 0.3 , and 0.6 ± 0.1 g C m⁻² (Figure 6), indicating that most of the CO₂ release or R_H and the increase in MBC came from the decomposition of the SOC in microbial models. The partial correlation analyses further showed a significant correlation between R_H and Δ MBC (*R* = 0.81, *P* < 0.01) in the model with MRF₃, and the changes of EBC in models with MRF₂ or MRF₃ were also positively correlated with R_H (*R* = 0.45, *P* < 0.01 and 0.48, *P* < 0.01 respectively).

The water response function also influenced the contribution of different soil C pools to R_H (Figures 5 and 6) through changing the model parameters. In conventional models, the most sensitive parameters were the decomposition rate of fast and slow C pools and the fractions of transfer between the two pools. Among these parameters, the decomposition rate of fast pools (k_f) and the fraction of C transferred from fast to slow pools (f_{fs}), and the initial value of fast pool size (f_f) in models with MRF₁ or MRF₂ scheme was higher than models with MRF₃ or MRF₄ scheme (Figure S3a). While the sensitivities for a fraction of C in slow C





Figure 6. The changes of soil heterotrophic respiration (R_H) and four different carbon pools in the microbial model (soil organic carbon, SOC; microbial biomass, MBC; dissolved organic carbon, DOC; enzyme biomass, EBC). The drawings at the top right show cumulative changes in different carbon pools (MRF₁-MRF₄ are moisture response functions from the model of CABLE, IBIS, LPJ, and TECO, respectively). TECO, Terrestrial ECOsystem.

pool transferring to fast C pool (f_{sf}), the decomposition rate of slow C pool (k_s) and the initial value of slow C pool size (f_s) in models with MRF₃ or MRF₄ were higher than models with MRF₁ or MRF₂ (Figure S3a). In microbial models, further analysis showed that the parameter of carbon use efficiency (CUE₀) is the most sensitive in all microbial models with MRF₁ to MRF₄ schemes, followed by parameters of the initial sizes of MBC, EBC, and DOC (Figure S3b). Besides, MRF affected the sensitivity of the loss rate of the enzyme (r_{EL}), maximum catalytic rate of SOC to DOC (V_{max0}), and the size of MBC and EBC (f_M and f_E) in microbial models (Figure S3b).

4. Discussions

4.1. Simulated Pulse Dynamics of Soil Heterotrophic Respiration by Different Models

Previous modeling studies have shown that there is controversy about the performance of linear conventional or nonlinear microbial models in simulating nonlinear soil C processes (Lawrence et al., 2009; Shi et al., 2018; Wieder et al., 2013). This study, however, shows that both the conventional and microbial models can capture a typical nonlinear C process (i.e., the R_H in response to rain pulse effect) in a semi-arid grassland (Figure 3). The simulations of the conventional model are even closer to the observed data than the microbial models in our study based on the results of the deviance information criterion (Figure 4).

Some laboratory-based experiments have suggested that the response of soil C to drying-rewetting events could be better explained by microbial models because they represented more realistic mechanisms of soil C-cycle processes (Manzoni et al., 2014, 2016; Salazar et al., 2018). There are great differences between laboratory experiments and field experiments (Poorter et al., 2016; Xu et al., 2019). Meanwhile, the development of microbial models must link micro-scale mechanisms to macro-scale models, which has to face several challenges compared to linear models, such as parameterization (Luo & Schurr, 2020; Wieder, Allison, et al., 2015). In our study, the microbial nonlinear model utilizes the enzyme biomass to modify available



substrate (DOC), and the microbial biomass and Michaelis-Menten kinetic to adjust the decomposition rate of soil C to match the nonlinear pulsed response in R_H (Figures 1 and 3). Although this type of microbial model characterizes nonlinear soil C processes more realistically than the conventional linear model, the latter has prevailed in simulations of linear or nonlinear soil C processes on regional or global scales over long periods due to the simpler C pool structure (Lawrence et al., 2009; Waring & Powers, 2016; Wieder, Allison, et al., 2015). The conventional linear model based on the CENTURY-like C pool structure (Jenkinson et al., 1991; Parton et al., 1987) describes the observed nonlinear response of soil C-released CO₂ well by incorporating appropriate environmental response parametrization schemes (e.g., soil moisture-response functions in this study) with optimized decomposition rate parameters (Figures 3 and 5). Other studies have also shown the conventional models performed well in simulating the nonlinear soil C processes. For example, Waring and Powers (2016) have reported that the modified linear model could reproduce the pulsed dynamics of R_H based on the field experiments in a tropical dry forest.

The development of microbial models is ongoing, and whether the linear conventional model can capture the nonlinear soil C processes as well as the microbial model remains debatable. We argue that these two types of models can both be improved to simulate nonlinear phenomena of soil C processes. Different model structures could be beneficial for unraveling the mechanisms underlying the different nonlinear soil C processes.

4.2. The Role of Moisture-Response Parameterization

In our study, we found that differences among the schemes of moisture-response function dominated the uncertainties of models in simulating the rain-pulsed dynamics of R_H (Figures 3 and 4). However, the empirical moisture-response functions vary widely among models (Lei et al., 2018; Sierra et al., 2015), and their values used to modify the decomposition processes are greatly different within the same variation range of soil moisture (Figure S1). This study highlights the importance of the moisture-response function in conventional or microbial models to better simulate the nonlinear soil C processes. The moisture-response function in conventional models affects the decomposition rates of different soil C pools and their response to environmental changes (Moyano et al., 2013), while in microbial models, the moisture-response function is generally considered to modify the function of Michaelis-Menten kinetics when microbial mechanisms are embedded in models (Abramoff et al., 2018; Wieder et al., 2013). Other factors can also affect the mechanistic relationship between the R_H and soil moisture, such as soil properties in pore-scale models (Moyano et al., 2012, 2013), microbial dormancy (Salazar et al., 2018), and soil water transfer processes (Manzoni et al., 2014, 2016; Sánchez-García, Doerr, et al., 2020; Sánchez-García, Oliveira, et al., 2020). Yan et al. (2018) have also considered primary physicochemical and biological processes to develop a novel moisture function of R_H .

This study further shows different moisture-response functions lead to great variability in the estimated values of some parameters among models (e.g., k_f in conventional models and CUE_0 in microbial models) (Figure S2). Such different parameters further cause different contributions of multiple C pools to R_H among models (Figures 5–6). As a result, the variation in predicting soil C dynamics is further increased among different models. Thus, it is crucial to use the data assimilation approach to constrain the parameters for different models. Meanwhile, more informative datasets should be used to improve model parameterization in simulating the soil C dynamics in response to environmental changes (Luo & Schurr, 2020; Shi et al., 2018; Wang et al., 2009; Xia et al., 2020).

4.3. The Implications for the Development of Soil Carbon Models

This study indicates that conventional linear models with optimized parameterizations (e.g., moisture-response functions) could reproduce non-linear response processes of soil C dynamics (e.g., pulse dynamic of R_H in grasslands) as well as microbial non-linear models do. This finding implies that parameterization schemes representing the mechanisms of soil C response to environmental factors are still very important for the development of both conventional linear or microbial models. According to previous studies, the uncertainty due to parameterizations of environmental response functions is comparable to that from external climate changes and biogeochemical nutrient limitation (Exbrayat et al., 2013; Moyano et al., 2013; Nishina



et al., 2014). Microbial models, which reflect more realistic non-linear soil C processes than conventional models, are one of the promising future developments in Earth system models (Luo et al., 2016; Manzoni et al., 2016; Schimel & Weintraub, 2003; Shi et al., 2018; Wieder et al., 2013). In this study, we highlight the importance of environmental response functions (e.g., moisture-response functions) in determining the uncertainty of soil C dynamics (Figure 4).

5. Conclusions

Our results show that linear conventional models can capture the pulse dynamics of soil heterotrophic respiration in a semi-arid grassland as well as nonlinear microbial models do. Parameterization with a suitable moisture-response function dominates the performances of these two types of models. This study implies that the capability of the current Earth system models in simulating nonlinear soil C phenomena can be improved by not only the incorporation of soil microbial dynamics but also by more accurate parameterizations of environmental impacts. In grassland ecosystems, as shown in this study, an improved parameterization of soil moisture functions is important in reducing the model uncertainty on soil C cycling. Thus, we recommend more in situ observations of nonlinear C phenomena for improving parameter estimates under alternative carbon-cycle model structures.

Data Availability Statement

We have archived the observational data in Figshare (doi.org/10.6084/m9.figshare.13114637).

References

- Abramoff, R., Xu, X., Hartman, M., O'Brien, S., Feng, W., Davidson, E., et al. (2018). The millennial model: In search of measurable pools and transformations for modeling soil carbon in the new century. *Biogeochemistry*, *137*(1–2), 51–71. https://doi.org/10.1007/s10533-017-0409-7
- Allison, S. D., Wallenstein, M. D., & Bradford, M. A. (2010). Soil-carbon response to warming dependent on microbial physiology. *Nature Geoscience*, *3*(5), 336–340. https://doi.org/10.1038/ngeo846
- Birch, H. F. (1964). Mineralization of plant nitrogen following alternate wet and dry conditions. *Plant and Soil*, 20, 43–49. https://doi. org/10.1007/bf01378096
- Bradford, M. A., Wieder, W. R., Bonan, G. B., Fierer, N., Raymond, P. A., & Crowther, T. W. (2016). Managing uncertainty in soil carbon feedbacks to climate change. *Nature Climate Change*, 6(8), 751–758. https://doi.org/10.1038/nclimate3071
- Byrne, K. A., & Kiely, G. (2006). Partitioning of respiration in an intensively managed grassland. *Plant and Soil*, 282(1–2), 281–289. https://doi.org/10.1007/s11104-005-6065-z
- Davidson, E. A., & Janssens, I. A. (2006). Temperature sensitivity of soil carbon decomposition and feedbacks to climate change. *Nature*, 440(7081).
- Doetterl, S., Stevens, A., Six, J., Merckx, R., Van Oost, K., Casanova Pinto, M., et al. (2015). Soil carbon storage controlled by interactions between geochemistry and climate. *Nature Geoscience*, 8(10), 780–783. https://doi.org/10.1038/ngeo2516
- Du, Z., Zhou, X., Shao, J., Yu, G., Wang, H., Zhai, D., et al. (2017). Quantifying uncertainties from additional nitrogen data and processes in a terrestrial ecosystem model with Bayesian probabilistic inversion. *Journal of Advances in Modeling Earth Systems*, 9(1), 548–565. https://doi.org/10.1002/2016ms000687
- Exbrayat, J.-F., Pitman, A. J., Zhang, Q., Abramowitz, G., & Wang, Y.-P. (2013). Examining soil carbon uncertainty in a global model: Response of microbial decomposition to temperature, moisture and nutrient limitation. *Biogeosciences*, *10*(11), 7095–7108. https://doi. org/10.5194/bg-10-7095-2013
- Falloon, P., Jones, C. D., Ades, M., & Paul, K. (2011). Direct soil moisture controls of future global soil carbon changes: An important source of uncertainty. *Global Biogeochemical Cycles*, 25(3), GB3010. https://doi.org/10.1029/2010gb003938
- Friedlingstein, P., Cox, P., Betts, R., Bopp, L., von Bloh, W., Brovkin, V., et al. (2006). Climate-carbon cycle feedback analysis: Results from the C4MIP model intercomparison. *Journal of Climate*, 19(14), 3337–3353. https://doi.org/10.1175/jcli3800.1
- German, D. P., Marcelo, K. R. B., Stone, M. M., & Allison, S. D. (2012). The Michaelis-Menten kinetics of soil extracellular enzymes in response to temperature: A cross-latitudinal study. Global Change Biology, 18(4), 1468–1479. https://doi.org/10.1111/j.1365-2486.2011.02615.x
- Hastings, W. K. (1970). Monte Carlo sampling methods using Markov chains and their applications. *Biometrika*, 57(1), 97–109. https://doi.org/10.1093/biomet/57.1.97
- Jenkinson, D. S., Adams, D. E., & Wild, A. (1991). Model estimates of CO₂ emissions from soil in response to global warming. *Nature*, 351(6324), 304–306. https://doi.org/10.1038/351304a0
- Jiang, J., Huang, Y., Ma, S., Stacy, M., Shi, Z., Ricciuto, D. M., et al. (2018). Forecasting responses of a Northern Peatland carbon cycle to elevated CO2 and a Gradient of experimental warming. *Journal of Geophysical Research: Biogeosciences*, 123(3), 1057–1071. https://doi. org/10.1002/2017jg004040
- Kucharik, C. J., Foley, J. A., Delire, C., Fisher, V. A., Coe, M. T., Lenters, J. D., et al. (2000). Testing the performance of a dynamic global ecosystem model: Water balance, carbon balance, and vegetation structure. *Global Biogeochemical Cycles*, *14*(3), 795–825. https://doi.org/10.1029/1999gb001138
- Lawrence, C. R., Neff, J. C., & Schimel, J. P. (2009). Does adding microbial mechanisms of decomposition improve soil organic matter models? A comparison of four models using data from a pulsed rewetting experiment. *Soil Biology and Biochemistry*, 41(9), 1923–1934. https://doi.org/10.1016/j.soilbio.2009.06.016

Acknowledgments

This work was supported by the National Key Research and Development Program of China (2017YFA0604603), National Natural Science Foundation of China (31800400 and 41630528), and the Fok Ying-Tong Education Foundation for Young Teachers in the Higher Education Institutions of China (grant 161016). We thank Yiqi Luo and Zhenggang Du for their suggestions and assistance with the data assimilation method. We also thank the editor and two anonymous reviewers for their constructive comments and suggestions.



Lei, L., Xia, J., Li, X., Huang, K., Zhang, A., Chen, S., et al. (2018). Water response of ecosystem respiration regulates future projection of net ecosystem productivity in a semiarid grassland. Agricultural and Forest Meteorology, 252, 175–191. https://doi.org/10.1016/j. agrformet.2018.01.020

Lenhart, T., Eckhardt, K., Fohrer, N., & Frede, H.-G. (2002). Comparison of two different approaches of sensitivity analysis, physics and chemistry of the Earth. Parts A/B/C, 27(9–10), 645–654. https://doi.org/10.1016/s1474-7065(02)00049-9

- Li, X., Zhu, Z., Zeng, H., & Piao, S. (2016). Estimation of gross primary production in China (1982-2010) with multiple ecosystem models. *Ecological Modelling*, 324, 33–44. https://doi.org/10.1016/j.ecolmodel.2015.12.019
- Liang, J., Zhou, Z., Huo, C., Shi, Z., Cole, J. R., Huang, L., et al. (2018). More replenishment than priming loss of soil organic carbon with additional carbon input. *Nature Communications*, 9(1), 3175. https://doi.org/10.1038/s41467-018-05667-7
- Liu, B., Wang, S. S., & Daoruna (2013). Effect of afforestation in sandy land on soil bulk density and particle-size composition in Duolun Count. Journal of Inner Mongolia Forestry Science & Technology, 39(2), 11–14.
- Luo, Y., Ahlströmg, A., Allison, S. D., Batjes, N. H., Brovkin, V., Carvalhais, N., et al (2016). Toward more realistic projections of soil carbon dynamics by Earth system models. *Global Biogeochemical Cycles*, 30(1), 40–56.
- Luo, Y., & Schuur, E. A. G. (2020). Model parameterization to represent processes at unresolved scales and changing properties of evolving systems. Global Change Biology, 26, 1109–1117. https://doi.org/10.1111/gcb.14939
- Ma, S., Jiang, J., Huang, Y., Shi, Z., Wilson, R. M., Ricciuto, D., et al. (2017). Data-constrained projections of methane fluxes in a Northern Minnesota Peatland in response to elevated CO₂ and warming. *Journal of Geophysical Research: Biogeosciences*, 122(11), 2841–2861. https://doi.org/10.1002/2017jg003932
- Manzoni, S., Moyano, F., Kätterer, T., & Schimel, J. (2016). Modeling coupled enzymatic and solute transport controls on decomposition in drying soils. Soil Biology and Biochemistry, 95, 275–287. https://doi.org/10.1016/j.soilbio.2016.01.006
- Manzoni, S., Schaeffer, S. M., Katul, G., Porporato, A., & Schimel, J. P. (2014). A theoretical analysis of microbial eco-physiological and diffusion limitations to carbon cycling in drying soils. Soil Biology and Biochemistry, 73, 69–83. https://doi.org/10.1016/j.soilbio.2014.02.008
- Metropolis, N., Rosenbluth, A. W., Rosenbluth, M. N., Teller, A. H., & Teller, E. (1953). Equation of state calculations by fast computing machines. *The Journal of Chemical Physics*, 21(6), 1087–1092. https://doi.org/10.1063/1.1699114
- Moyano, F. E., Manzoni, S., & Chenu, C. (2013). Responses of soil heterotrophic respiration to moisture availability: An exploration of processes and models. Soil Biology and Biochemistry, 59, 72–85. https://doi.org/10.1016/j.soilbio.2013.01.002
- Moyano, F. E., Vasilyeva, N., Bouckaert, L., Cook, F., Craine, J., Curiel Yuste, J., et al. (2012). The moisture response of soil heterotrophic respiration: Interaction with soil properties. *Biogeosciences*, 9(3), 1173–1182. https://doi.org/10.5194/bg-9-1173-2012
- Nishina, K., Ito, A., Beerling, D. J., Cadule, P., Ciais, P., Clark, D. B., et al. (2014). Quantifying uncertainties in soil carbon responses to changes in global mean temperature and precipitation. *Earth System Dynamics*, 5(1), 197–209. https://doi.org/10.5194/esd-5-197-2014
- Parton, W. J., Schimel, D. S., Cole, C. V., & Ojima, D. S. (1987). Analysis of factors controlling soil organic matter levels in Great Plains grasslands. Soil Science Society of America Journal, 51(5), 1173–1179. https://doi.org/10.2136/sssaj1987.03615995005100050015x
- Poorter, H., Fiorani, F., Pieruschka, R., Wojciechowski, T., Putten, W. H., Kleyer, M., et al. (2016). Pampered inside, pestered outside? Differences and similarities between plants growing in controlled conditions and in the field. *New Phytologist*, 212(4), 838–855. https:// doi.org/10.1111/nph.14243
- Salazar, A., Sulman, B. N., & Dukes, J. S. (2018). Microbial dormancy promotes microbial biomass and respiration across pulses of drying-wetting stress. Soil Biology and Biochemistry, 116, 237–244. https://doi.org/10.1016/j.soilbio.2017.10.017
- Sánchez-García, C., Doerr, S. H., & Urbanek, E. (2020). The effect of water repellency on the short-term release of CO2 upon soil wetting. Geoderma, 375, 114481. https://doi.org/10.1016/j.geoderma.2020.114481
- Sánchez-García, C., Oliveira, B. R., Keizer, J. J., Doerr, S. H., & Urbanek, E. (2020). Water repellency reduces soil CO2 efflux upon rewetting. Science of the Total Environment, 708, 135014. https://doi.org/10.1016/j.scitotenv.2019.135014
- Schimel, J., & Weintraub, M. N. (2003). The implications of exoenzyme activity on microbial carbon and nitrogen limitation in soil: A theoretical model. Soil Biology and Biochemistry, 35(4), 549–563. https://doi.org/10.1016/s0038-0717(03)00015-4
- Shi, Z., Crowell, S., Luo, Y., & Moore, B., 3rd (2018). Model structures amplify uncertainty in predicted soil carbon responses to climate change. Nature Communications, 9(1), 2171. https://doi.org/10.1038/s41467-018-04526-9
- Sierra, C. A., Trumbore, S. E., Davidson, E. A., Vicca, S., & Janssens, I. (2015). Sensitivity of decomposition rates of soil organic matter with respect to simultaneous changes in temperature and moisture. *Journal of Advances in Modeling Earth Systems*, 7(1), 335–356. https:// doi.org/10.1002/2014ms000358
- Sitch, S., Smith, B., Prentice, I. C., Arneth, A., Bondeau, A., Cramer, W., et al. (2003). Evaluation of ecosystem dynamics, plant geography and terrestrial carbon cycling in the LPJ dynamic global vegetation model. *Global Change Biology*, 9(2), 161–185. https://doi. org/10.1046/j.1365-2486.2003.00569.x
- Spiegelhalter, D. J., Best, N. G., Carlin, B. P., & van der Linde, A. (2002). Bayesian measures of model complexity and fit. *Journal of the Royal Statistical Society B*, 64(4), 583–639. https://doi.org/10.1111/1467-9868.00353
- Sponseller, R. A. (2007). Precipitation pulses and soil CO₂ flux in a Sonoran Desert ecosystem. *Global Change Biology*, 13(2), 426–436. https://doi.org/10.1111/j.1365-2486.2006.01307.x
- Sulman, B. N., Moore, J. A. M., Abramoff, R., Averill, C., Kivlin, S., Georgiou, K., et al. (2018). Multiple models and experiments underscore large uncertainty in soil carbon dynamics. *Biogeochemistry*, 141(2), 109–123. https://doi.org/10.1007/s10533-018-0509-z
- Todd-Brown, K. E. O., Randerson, J. T., Post, W. M., Hoffman, F. M., Tarnocai, C., Schuur, E. A. G., & Allison, S. D. (2013). Causes of variation in soil carbon simulations from CMIP5 Earth system models and comparison with observations. *Biogeosciences*, 10(3), 1717–1736. https://doi.org/10.5194/bg-10-1717-2013
- Wang, G., Post, W. M., & Mayes, M. A. (2013). Development of microbial-enzyme-mediated decomposition model parameters through steady-state and dynamic analyses. *Ecological Applications*, 23(1), 255–272. https://doi.org/10.1890/12-0681.1
- Wang, Y. P., Jiang, J., Chen-Charpentier, B., Agusto, F. B., Hastings, A., Hoffman, F., et al. (2016). Responses of two nonlinear microbial models to warming and increased carbon input. *Biogeosciences*, 13(4), 887–902. https://doi.org/10.5194/bg-13-887-2016
- Wang, Y. P., Law, R. M., & Pak, B. (2010). A global model of carbon, nitrogen and phosphorus cycles for the terrestrial biosphere. *Biogeosciences*, 7(7), 2261–2282. https://doi.org/10.5194/bg-7-2261-2010
- Wang, Y.-P., Trudinger, C. M., & Enting, I. G. (2009). A review of applications of model-data fusion to studies of terrestrial carbon fluxes at different scales. Agricultural and Forest Meteorology, 149(11), 1829–1842. https://doi.org/10.1016/j.agrformet.2009.07.009
- Waring, B. G., & Powers, J. S. (2016). Unraveling the mechanisms underlying pulse dynamics of soil respiration in tropical dry forests. *Environmental Research Letters*, 11(10).
- Weng, E., & Luo, Y. (2008). Soil hydrological properties regulate grassland ecosystem responses to multifactor global change: A modeling analysis, Journal of Geophysical Research, 113(G3).



- Wieder, W. R., Allison, S. D., Davidson, E. A., Georgiou, K., Hararuk, O., He, Y., et al. (2015). Explicitly representing soil microbial processes in Earth system models. *Global Biogeochemical Cycles*, 29(10), 1782–1800. https://doi.org/10.1002/2015gb005188
- Wieder, W. R., Bonan, G. B., & Allison, S. D. (2013). Global soil carbon projections are improved by modeling microbial processes. Nature Climate Change, 3(10), 909–912. https://doi.org/10.1038/nclimate1951
- Wieder, W. R., Grandy, A. S., Kallenbach, C. M., Taylor, P. G., & Bonan, G. B. (2015). Representing life in the Earth system with soil microbial functional traits in the MIMICS model. *Geoscientific Model Development*, 8(6), 1789–1808. https://doi.org/10.5194/gmd-8-1789-2015
 Xia, J., Luo, Y., WangWang, Y.-P., & Hararuk, O. (2013). Traceable components of terrestrial carbon storage capacity in biogeochemical
- models. *Global Change Biology*, 19, 2104–2116. https://doi.org/10.1111/gcb.12172 Xia, J., Wang, J., & Niu, S. (2020). Research challenges and opportunities for using big data in global change biology. *Global Change Biology*,
- 26, 6040–6061. https://doi.org/10.1111/gb.15317
- Xu, X., Yan, L., & Xia, J. (2019). A threefold difference in plant growth response to nitrogen addition between the laboratory and field experiments. *Ecosphere*, *10*(1), e02572. https://doi.org/10.1002/ecs2.2572
- Yan, L., Chen, S., Xia, J., & Luo, Y. (2014). Precipitation regime shift enhanced the rain pulse effect on soil respiration in a semi-arid steppe. *PloS One*, 9(8), e104217. https://doi.org/10.1371/journal.pone.0104217
- Yan, Z., Bond-Lamberty, B., Todd-Brown, K. E., Bailey, V. L., Li, S., Liu, C., & Liu, C. (2018). A moisture function of soil heterotrophic respiration that incorporates microscale processes. *Nature Communications*, 9(1), 2562. https://doi.org/10.1038/s41467-018-04971-6
- Yuan, Z.-Y., Li, L.-H., Han, X.-G., Huang, J.-H., Jiang, G.-M., & Wan, S.-Q. (2005). Soil characteristics and nitrogen resorption in Stipa krylovii native to northern China. *Plant and Soil*, 273(1–2), 257–268. https://doi.org/10.1007/s11104-004-7941-7
- Zhang, W. L., Chen, S. P., Chen, J., Wei, L., Han, X. G., & Lin, G. H. (2007). Biophysical regulations of carbon fluxes of a steppe and a cultivated cropland in semiarid Inner Mongolia. *Agricultural and Forest Meteorology*, *146*(3–4), 216–229. https://doi.org/10.1016/j. agrformet.2007.06.002
- Zhou, J., Xia, J., Wei, N., Liu, Y., Bian, C., Bai, Y., & Luo, Y. (2021). A traceability analysis system for model evaluation on land carbon dynamics: Design and applications. *Ecological Processes*, 10, 12. https://doi.org/10.1186/s13717-021-00281-w